Author's response to reviews

Title: Effect of educational program for physical activity using pedometer on anthropometric measurements after childbirth, A randomized controlled trial

Authors:

Masumeh Mrs SechinMaturi (nastaran47@yahoo.com)
Pourandokht Mrs Afshary (p.afshary@yahoo.com)
Parvin Dr Abedi (parvinabedi@ymail.com)

Version: 2 Date: 2 August 2011

Author's response to reviews: see over
Dear Editor

I made changes you requested in the abstract, writing the trial registration number, Authors’ contributions and competing interests.

Kind Regards

Dr Parvin Abedi